Author's response to reviews

Title: Impact of an online writing aid tool for writing a randomized trial report: The COBWEB (COnsort Based WEB tool) randomized controlled trial

Authors:

Caroline Barnes (caroline.grace.barnes@gmail.com)
Isabelle Boutron (isabelle.boutron@htd.aphp.fr)
Bruno Giraudeau (bruno.giraudeau@univ-tours.fr)
Raphael Porcher (raphael.porcher@htd.aphp.fr)
Douglas Altman (doug.altman@cs.m.ox.ac.uk)
Philippe Ravaud (philippe.ravaud@htd.aphp.fr)

Version: 4 Date: 21 August 2015

Author's response to reviews:

Dear Claire Barnard

Please find in attached the revised manuscript. We now clarified that all parts of the trial were conducted in Paris and that all students were based in Paris.

best wishes
Isabelle Boutron